Author’s response to reviews

Title: Role of the EGF +61A>G polymorphism in melanoma pathogenesis: an experience on a large series of Italian cases and controls.

Authors:

Milena Casula (casulam@yahoo.it)
Mauro Alaibac (mauro.alaiabac@unipd.it)
Maria A Pizzichetta (pizzichetta@cro.it)
Riccardo Bono (r.bono@idi.it)
Paolo A Ascierto (pasciert@tin.it)
Ignazio Stanganelli (igstanga@tin.it)
Sergio Canzanella (househospital@tin.it)
Grazia Palomba (graziap68@yahoo.it)
Edoardo Zattra (edoardo.zattra@unipd.it)
Giuseppe Palmieri (gpalmieri@yahoo.com)

Version: 3 Date: 6 July 2009

Author’s response to reviews:

Dear Editor,

Please find enclosed the revised manuscript by Casula et al., entitled “Role of the EGF +61A>G polymorphism in melanoma pathogenesis: an experience on a large series of Italian cases and controls” (MS# 1673532550266316).

Revision of the manuscript was based on the Reviewer 1 remarks. Changes have been highlighted (in yellow) into the paper and here listed point-by-point:

1. Individual headings were given for Table 1A-B-C, which have now become Tables 1, 2, and 3 (respectively); changes have been reported into the text also.
2. Odds ratio (along with 95%CI) for allele frequencies has been added into Tables 2 and 3 (1B and 1C of the old version).

Hoping to have addressed these additional issues and looking forward to receiving good news from You.

Sincerely Yours,

Giuseppe Palmieri